Author's response to reviews

Title: Factors influencing patient satisfaction with dental appearance and treatments they desire to improve aesthetics

Authors:

Mon Mon Tin-Oo (monmonto@kb.usm.my)
Norkhafizah Saddki (fizah@kb.usm.my)
Nurhidayati Hassan (ida86_kun@yahoo.com)

Version: 5 Date: 19 February 2011

Author's response to reviews:

To,
The BMC Series Editorial Production Team

Dear Sir,

Thank you very much for the message.

I have submitted the revised manuscripts with changes made for major revision. Box1 was changed to table format and titled it as Table 1. References to Box1 and the rest of the table numbers were also updated within the text.

Previous Table 1 changed to Table 2,
Table 2 changed to Table 3
Table 3 changed to Table 4
Table 4 changed to Table 5
Table 5 changed to Table 6
Table 6 changed to Table 7

Since I will be away from the station I can be able to communicate after 23 February 2010.

Thank you

Mon Mon Tin-Oo